Author's response to reviews

Title: A prospective study of monitoring practices for metabolic disease in antipsychotic-treated community psychiatric patients

Authors:

Paul Mackin (paul.mackin@ncl.ac.uk)
David R Bishop (d.r.bishop@ncl.ac.uk)
Helen MO Watkinson (h.m.o.watkinson@ncl.ac.uk)

Version: 3  Date: 14 June 2007

Author's response to reviews:

Dear Editorial Production Team

The formatting changes you requested have been made to our manuscript.

1. Qualifications of authors have been removed.
2. Our affiliation has been written in full and each author is linked to their affiliation with a superscript symbol.
3. The corresponding author has been identified with a superscript symbol.
4. The word count has been removed.
5. Sentence case has been used for all headings and sub-headings.
6. The blank page has been removed.
7. Table placement text has been removed
8. A conclusions section has been added.
9. Tables appear in the order in which they are referred to in the text.

I hope these changes are satisfactory.

Yours faithfully

Paul Mackin